Author’s response to reviews

Title: Using technology to support caregivers of people living with dementia: A systematic review of research output and quality

Authors:

Amy Waller (amy.waller@newcastle.edu.au)
Sophie Dilworth (sophie.dilworth@newcastle.edu.au)
Elise Mansfield (elise.mansfield@newcastle.edu.au)
Rob Sanson-Fisher (rob.sanson-fisher@newcastle.edu.au)

Version: 2 Date: 22 Sep 2017

Author’s response to reviews:

Dear Dr Ya-Huei Wu,

Thank you for recently reviewing our manuscript entitled: Using technology to support caregivers of people living with dementia: A systematic review of research output and quality (BGTC-D-17-00345). Enclosed is a revised version of our manuscript, addressing the points identified by the Reviewers as outlined below.

Table 1: The missing references have been added to Table 1.

We look forward to hearing back from you soon regarding the status of this manuscript.

Yours Sincerely,

Dr Amy Waller (on behalf of the authors)